Author's response to reviews

Title: Perimenopausal contraception in Turkish women: A cross-sectional study

Authors:

Nevin H Sahin (nevinsahin34@yahoo.com)
Sema B Kharbouch (semaca80@yahoo.com)

Version: 11  Date: 7 March 2007

Author's response to reviews:

We have made the necessary changes in the manuscript format and revised the paper according to "Core pre-accept"

Nevin H. Sahin, Ph D
Istanbul University, Florence Nightingale School of Nursing, Department of Obstetric & Gynaecologic Nursing
Istanbul, Turkey